



RESEARCH

Received 29 August 2023 Accepted 11 December 2023 Available online 11 December 2023 Version of Record published 16 January 2024

Pituitary dysfunction following mild traumatic brain injury in female athletes

Lára Ósk Eggertsdóttir Claessen^{®1,2}, Hafrún Kristjánsdóttir³, María Kristín Jónsdóttir^{4,5}, Sigrún Helga Lund^{6,7}, Ingunn Unnsteinsdóttir Kristensen⁵ and Helga Ágústa Sigurjónsdóttir^{1,8}

¹Faculty of Medicine, School of Health Sciences, University of Iceland, Reykjavik, Iceland

²Department of Emergency Medicine, Landspitali – The National University Hospital of Iceland, Reykjavik, Iceland

³Physical Activity, Physical Education, Sport, and Health (PAPESH) Research Centre, Sports Science Department, School of Social Sciences, Reykjavik University, Reykjavik, Iceland

⁴Mental Health Services, Landspitali – The National University Hospital of Iceland, Reykjavik, Iceland

⁵Department of Psychology, School of Social Sciences, Reykjavik University, Reykjavik, Iceland

6deCODE Genetics, Inc/Amgen Inc., Reykjavik, Iceland

⁷School of Engineering and Natural Sciences, University of Iceland, Reykjavik, Iceland

⁸Department of Medicine, Landspitali – The National University Hospital of Iceland, Reykjavik, Iceland

Correspondence should be addressed to L Ó Eggertsdóttir Claessen: laraclaessen@gmail.com

Abstract

Objective: Pituitary dysfunction following mild traumatic brain injury can have serious physical and psychological consequences, making correct diagnosis and treatment essential. To the best of our knowledge, this study is the first to study the prevalence of pituitary dysfunction following mild traumatic brain injury in an all-female population following detailed endocrinological work-up after screening for pituitary dysfunction in female athletes.

Design: This is a retrospective cohort study.

Methods: Hormone screening blood tests, including serum blood values for thyroid-stimulating hormone, free thyroxin, insulin-like growth factor 1, prolactin, cortisol, follicle-stimulating hormone, luteinizing hormone, estrogen and progesterone, were taken in 133 female athletes. Results were repeatedly outside the reference value in 88 women necessitating further endocrinological evaluation. Two of those were lost to follow-up, and further endocrinological evaluation was performed in 86 participants.

Results: Six women (4.6%, n = 131) were diagnosed with hypopituitarism, four (3.1%) with central hypothyroidism and two with growth hormone deficiency (1.5%). Ten women (7.6%) had hyperprolactinemia, and four (3.1%) of them had prolactinoma. Medical treatment was initiated in 13 (9.9%) women. Significant prognostic factors were not found.

Conclusions: As 12.2% of female athletes with a history of mild traumatic brain injury had pituitary dysfunction (hypopituitarism 4.6%, hyperprolactinemia 7.6%), we conclude that pituitary dysfunction is an important consideration in post-concussion care. Hyperprolactinemia in the absence of prolactinoma may represent pituitary or hypothalamic injury following mild traumatic brain injury.

Significance statement

Mild traumatic brain injury (mTBI) has become a growing public health concern as 50 million people worldwide sustain a traumatic brain injury annually, with mTBI being the most common (70–90%). As studies on mTBI have



focused on mostly male populations this study aims to explore pituitary dysfunction (PD) in female athletes following mTBI. To the best of our knowledge, it is the first all-female study on PD following mTBI.

The study found that 12.2% of the participating women had PD after mTBI. Six (4.6%) had hypopituitarism and ten (7.6%) had hyperprolactinemia. These findings suggest that PD following mTBI is an important consideration that endocrinologists and other medical staff working with athletes need to be aware of.

Keywords: hypopituitarism (HP); mild traumatic brain injury (mTBI); traumatic brain injury (TBI); sport-related concussion (SRC); female athletes; hyperprolactinemia

Introduction

Traumatic brain injury (TBI) is divided into mild (mTBI), moderate (moTBI), and severe (sTBI) injury. The most common is mTBI (1) which is caused by mechanical force being transmitted to the brain by a blow to the head. neck or body, acceleration-deceleration movement, or forces from a blast injury. Symptoms can present immediately, within hours or days and may or may not include loss of consciousness for less than 30 min (2, 3). Furthermore, recent diagnostic criteria for mTBI include a Glasgow coma scale (GCS) score of 13–15 after 30 min from the head injury, post-traumatic amnesia for less than 24 h, and normal neuroimaging studies (3, 4, 5). Symptoms of mTBI often resolve within a few weeks although prolonged cognitive and psychological effects can occur (6, 7). Furthermore, it has been demonstrated that mTBI can lead to hypopituitarism (HP) with a prevalence of 13-48% (8, 9, 10, 11, 12, 13, 14, 15), making it an important consideration of post-concussion care.

It has been speculated that HP following TBI may be due to direct or indirect injury to the pituitary gland or hypothalamus with four possible mechanisms of indirect injury (16):

- 1. Vascular injury to arteries supplying the pituitary gland resulting in ischemic damage (17).
- 2. Neuroinflammation and cytokine release can occur following mTBI causing pituitary damage (17).
- 3. Autoimmunity may also have a role in indirect pituitary injury following mTBI, as studies have found anti-pituitary antibodies in patients with previous mTBI (18).
- 4. Uncontrolled release of excitatory neurotransmitters following injury may damage neuronal cells in the pituitary gland by affecting cellular permeability (19).

Undiagnosed HP following mTBI can have serious physical and psychological consequences depending on which axes are affected (20, 21). While untreated glucocorticoid deficiency (GCD) can be life-threatening, symptoms of growth hormone deficiency (GHD) may be subtle and overlaps with symptoms of mTBI causing diagnostic delay. However, if left untreated, GHD can lead to decreased quality of life (22), metabolic alterations (23, 24, 25), osteopenia and osteoporosis (23, 25), and increased risk of cardiovascular and cerebrovascular morbidity and mortality (24, 25, 26). As hormonal supplementation therapy (HST) can reverse the metabolic and psychological effects of GHD (23, 25) accurate diagnosis and treatment is vital. Furthermore, even subclinical untreated hypothyroidism can have serious effects such as increased cardiovascular morbidity, and mortality (27) and gonadotropic deficiency can impact fertility and cause comorbidities of decreased gonadal hormones. Earlier reports indicate that the somatotropic axis is most commonly affected by mTBI with a prevalence of 8-48% (8, 11, 12, 13, 14, 15, 28, 29). The somatotroph cells may be prone to vascular injury as they receive blood supply via the long hypophysial portal vessels that are especially vulnerable (30).

Contrary to other pituitary hormones, which can become deficient following mTBI, hyperprolactinemia (HPRL) can occur following TBI (mTBI, moTBI and sTBI) (15, 31, 32, 33, 34, 35, 36). Pituitary gland injury affects the dopaminergic inhibitory control of prolactin release, resulting in rising serum prolactin (s-prolactin) levels (30). Thus, HPRL may be a sign of pituitary or hypothalamic injury following TBI (36) and may be a marker of TBI severity, as it has been shown to correlate negatively with GCS (37).

Definite diagnosis of GCD or GHD involves stimulation testing, the synacthen test for GCD diagnosis (38) and the insulin tolerance test (ITT) for GHD and GCD diagnosis. Due to potential risks related to the ITT, a safer method such as the combined test with growth hormone releasing hormone and arginine (GHRH– arginine test) has been validated for GHD diagnosis (39, 40). As stimulation tests can be time consuming and expensive, a biochemical marker for GHD screening would be beneficial. Serum insulin-like growth factor 1 (s-IGF1) has been proposed as a screening tool for GHD as low s-IGF1 may indicate GHD. However, its use remains debatable as s-IGF1 within normal range does not exclude GHD (28, 40).

Female athletes remain an understudied population with regards to mTBI (41) and HP following mTBI even though they appear to be more susceptible to mTBI

(42) and recovery time seems longer than with male athletes (43). The aim of this study was to explore pituitary dysfunction (PD) including HP in female athletes following mTBI in sport and, to the best of our knowledge, it is the first study to do so.

Materials and methods

Study design and subjects

This study is a part of more extensive research on female athletes. A comprehensive description of the population inclusion criteria has been published (44, 45, 46). The study included women aged 18-45 years currently active in or retired from soccer, handball, basketball, ice hockey and martial arts in Iceland. Of the 508 women included in part 1 of the study, 166 women accepted further participation in part 2 including a detailed psychological interview focusing on mTBI history and neuropsychological testing (44, 46). Following part 2, all 166 women were invited to participate in part 3 of the study, presented here, with 151 women (91.0%, n = 166) accepting, 15 women (9.0%) were lost to follow-up. In part 3, a physical examination was conducted by the same medical doctor (LÓEC) participants including a neurological for all examination, and hormonal screening blood tests (SBT) for possible PD. Hormonal evaluation of all pituitary axes was performed with SBT taken in 133 women (88.1%, n=151), nine were lost to follow-up and nine were pregnant. All SBT results were reviewed by the same endocrinologist (HÁS). If SBT results were outside reference value (O-RV) in two or three repeated blood tests for each serum hormonal measurement, they were defined as abnormal (45).

Of the 133 women who had SBT taken, 88 women (66.2%, n=131) had results repeatedly O-RV necessitating further evaluation including a medical interview with an endocrinologist, physical examination and possibly further endocrinological testing (Fig. 1). Two of these 88 participants did not attend the visit despite repeated attempts to contact them and were thus lost to follow-up (Fig. 1). Thus, 86 women (64.6%, n=131) attended the medical interview with an endocrinologist followed by detailed endocrinological testing as indicated. All female participants diagnosed with PD requiring treatment or follow-up will be followed by the endocrinologist (HÁS).

Measurements and analytical methods

The SBT were taken at 08:00 h at the earliest convenient day for the participants and included serum thyroidstimulating hormone (s-TSH), serum free thyroxin (s-fT4), s-IGF1, s-prolactin, serum cortisol (s-cortisol), serum follicle-stimulating hormone (s-FSH), serum luteinizing hormone (s-LH), serum estrogen (s-estrogen) and serum progesterone (s-progesterone). If the first SBT was O-RV for each hormone (Table 1), the blood tests were repeated for reevaluation.

In the endocrinological interview, further information regarding mTBI history was gathered as well as information regarding previous medical history, medications and possible clinical symptoms of mTBI or HP. Thus, information on mTBI symptoms was gathered at four different times during the study period (part 1, part 2 and twice in part 3). Height, weight, blood pressure and heart rate were measured. Further endocrinological tests were then requested as necessary.

Endocrinological tests

When s-fT4 was below reference value (RV) (Table 1) serum levels of anti-thyroid peroxidase antibodies (s-anti-TPO) were measured using the EliA method (fluoroenzymeimmunoassay) to exclude autoimmune hypothyroidism. Central hypothyroidism was suspected if s-anti-TPO was negative along with low or normal TSH levels and low s-fT4 levels. Consequently, magnetic resonance imaging (MRI) was requested for further work-up.

For s-IGF1, results below median RV (Table 1) were considered abnormal if clinical symptoms of GHD such as decreased vitality and energy, impaired psychological well-being (22, 25), and changes in memory and attention (47, 48) were also present and further evaluation with a GHRH–arginine test was performed as described in the consensus guidelines (40). If s-IGF1 was repeatedly below median RV without any symptoms of GHD, further endocrinological evaluation was not performed.

An ITT was performed in one woman due to practical reasons and strong clinical symptoms indicating GHD. It was performed as described in the consensus guidelines (40).

The lactotroph axis was evaluated using s-prolactin measurements. When s-prolactin level was found to be elevated, a macroprolactin analysis was performed to differentiate between monomeric prolactin and macroprolactin. When s-prolactin was repeatedly above RV (Table 1), an MRI was performed.

When s-cortisol was below 350 nmol/L (49), plasma adrenocorticotropic hormone (ACTH) was measured and a high dose (250 µg) synacthen test was performed as earlier described (50, 51). A normal response was defined as s-cortisol \geq 440 nmol/L after either 30 or 60 min.

Ethics

The study was approved by the National Bioethics Committee (no. VSN-18-091), the Icelandic Data



Figure 1

Overview of the study design and the detailed endocrinological work-up. *As two of the 88 women who needed further endocrinological evaluation did not attend the medical interview with the endocrinologist they were excluded from the study. Thus, 131 women instead of 133 had SBT and 86 women instead of 88 had a detailed endocrinological evaluation. **GHRH-arginine tests were not performed for 19 women who did not have symptoms indicating possible GHD. ***Lost to follow-up (n = 1), pregnant when the test was to be conducted (n = 2) and one woman is being treated for hyperprolactinemia and followed by the endocrinologist before the GHRH-arginine stimulation test can be performed if necessary. \rightarrow Blood test results within RV; \uparrow Blood test results above RV; \downarrow Blood test results below RV. ACTH, adrenocorticotropic hormone; anti-TPO, anti-thyroid peroxidase antibodies; GHRH-arginine, growth hormone-releasing hormone and arginine; ITT, insulin tolerance test; MRI, magnetic resonance imaging; S-cortisol, serum cortisol; S-IGF1, serum insulin like growth factor 1.

Protection Authority, the Institutional Research Committee of Landspitali National University Hospital, Iceland, the chief medical officer of Landspitali National University Hospital, Iceland and Laeknasetrid outpatient clinic (OB/ei Tilv. 16).

Statistical analysis

Statistical analysis was performed using R (version 3.6.1). A two-sample *t*-test and the chi-squared test were used to compare women with PD and women with normal pituitary function to identify possible risk factors for PD (Table 2). Categorical data were examined for association significance using Fisher's exact test. The effect size (Table 2) was calculated using Cohen's *d* for the two-sample *t*-test, Phi (ϕ) for chi-squared test with 2 × 2 contingency tables. Population size needed for 80% power was calculated using Lehr's formula.

Results

Of the 131 women who had SBT taken, 86 (65.6%, n=131) had results repeatedly O-RV and were referred for further endocrinological evaluation. Following detailed endocrinological testing 16 women were diagnosed with PD (12.2%, n=131), 115 had normal pituitary function. Population characteristics and comparison between the two groups are presented in Table 2 (see also Fig. 1).

Thyroid hormone levels were O-RV in 15 participants (11.5%, n=131) and further work-up of the thyroid axis was performed in 12 (9.1%) of them (Fig. 1). Two of the 12 women had s-anti-TPO levels above RV and were diagnosed with autoimmune hypothyroidism. Six of the 10 women with s-anti-TPO within RV were not evaluated further as their thyroid hormone levels normalized during follow-up. Four women (3.1%,

					CV%				Spe	cimen
				Low	Med.	High				
Analyte	Assay name	Manufacturer	Instrument	contr	contr	contr	Reportable range	Reference value	Type	Storage
SH	TSH	Roche	Elecsys	11.1	1.3 1.3	2.2	0.005-100 mIU/L	0.270-4.20 mIU/L	Serum	+2-8°C
T4	FT4 III	Roche	Elecsys	4.3	1.5	2.6	0.5–100 pmol/L	12–22 pmol/L	Serum	+2-8°C
Prolactin	Prolactin II	Roche	Elecsys	1.5	3.0	2.4	0.0470–470 µg/L	Female (not pregnant) 4.79–23.3 μg/L	Serum	+2-8°C
Cortisol	Cortisol II	Roche	Elecsys	5.4	1.5	1.6	1.5–1750 nmol/L	Morning (06:10 h): 133–537 nmol/L, afternoon (16:08 h): 68.2–327 nmol/L	Serum	+2-8°C
GF1	IGF-I	Siemens	Immulite 2000	6.3	3.1	2.5	15–1000 μg/L	Manufacturer's age-dependent reference rangeª	Serum	+2-8°C or -20°C
ACTH	ACTH	Roche	Elecsys	2.7	0.6	0.7	1.0–2000 ng/L	7.2–63.3 ng/L	Plasma	+2-8°C
H	Growth hormone	Siemens	Immulite 2000	6.5	5.5	6.6	0.05-40 ng/mL	Females: up to 8 ng/mL	Serum	+2-8°C
Anti-TPO	EliA anti-TPO	Thermo Fisher	Phadia™ 250	I	I	I	I	<25 IU/mL = negative, 25–35 IU/ mL = equivocal, >35 IU/mL = positive	Serum	+2-8°C
Immulite p	roduct booklet. IG	F-1. Immulite 2000	IGF-1 (PIL2KIGF-4, 20	18-07-02).						

Table 1Hormone measurements and analytical methods of serum hormones measured.

hormone; Anti-TPO, anti-thyroid peroxidase antibodies; CV, coefficient of variation; fT4, free thyroxine; GH, growth hormone; High contr, high control; IGF1, insulin-like growth factor 1; Low contr, low control; Med. contr, median control; TSH, thyroid-stimulating hormone ACTH, adrenocorticotropic

n=131) were suspected having central hypothyroidism as they did not have s-anti-TPO and their s-TSH were low to normal with low fT4 levels. During follow-up of these four women, thyroid function tests were repeated at an interval ranging from 1 to 9 months with a mean of 3.2 months. Three of these four women had normal MRI results. One woman did not attend MRI and was lost to follow-up despite repeated attempts to contact her.

Of the 74 (56.5%, n=131) women with s-IGF1 levels in SBT below median RV (Fig. 1), 19 did not have clinical symptoms indicating GHD and were not evaluated further. Clinical symptoms of GHD were present in 53 participants who were referred to further endocrinological testing (ITT n=1, GHRH–arginine test n = 52). A GHRH–arginine test was completed in 48 of the 53 women (90.6%) (Fig. 1).

Elevated s-prolactin was found in 30 women (22.9%, n=131) (Fig. 1). Ten were taking medications that can cause HPRL (selective serotonin reuptake inhibitors (SSRI) n=2, hormone contraception n=5, SSRI and hormone contraception n=3) (52). Thus, further work-up for HPRL was not indicated in those women. Of the remaining 20 women who had elevated levels of s-prolactin, 18 had an MRI of the pituitary gland and two were lost to follow-up (Fig. 1). Seven of the 18 women had visible changes of the pituitary gland (hypopituitary atrophy and signs of a regressing prolactinoma n=1, microadenoma n=2, cystic/hemorrhagic adenoma n=1, concentric enlargement of the adenohypophysis with no visible tumor n=1, concentric enlargement of the pituitary gland n=1, arachnoid cyst n=1) and 11 had normal MRI results (Fig. 2). Four (3.1%, n=131) of the seven women with abnormal MRI results had a prolactinoma (hypopituitary atrophy and signs of a regressing prolactinoma n=1, microadenoma n=2, cystic/hemorrhagic adenoma n=1) (Fig. 2). In summary, of the 18 women who had further work-up of the lactotroph axis, 10 women (7.6%, n=131) had repeatedly elevated levels of s-prolactin and were diagnosed with HPRL. The s-prolactin levels normalized during follow-up in eight women who were not diagnosed with HPRL (Fig. 2). Four of the 10 women with HPRL were diagnosed with a prolactinoma and 6 women were not.

S-cortisol was below RV (below 350 mmol/L) in eight women (6.2%) (Fig. 1). Six were evaluated further with plasma ACTH measurements that were all within the RV. Five women needed further work-up with a synacthen test which was normal (peak s-cortisol \geq 440 nmol/L) for all.

As has been reported, one woman had gonadotropin levels below RV in SBT. However, further endocrinological evaluation was not indicated as she was taking hormonal contraception (HOC) (45).

Following a detailed endocrinological work-up, 16 (12.2%) of the 131 participating women had PD. Six women (4.6%) had HP (GHD n=2, 1.5% and **Table 2** Demographic and clinical characteristics of the study population. Women with pituitary dysfunction were compared to women with normal pituitary to identify possible risk factors for pituitary dysfunction. Statistical comparison between the two groups was not performed for sport, previous medical history, previous history of hormonal disease, hormonal contraception or menstrual changes as there were too few participants with PD for statistical analysis with the chi-square test. For the entire population (*n* = 131) the number of mTBI ranged from 1.0 to 4.0, BMI ranged from 19.1 to 46.5, time that passed from mTBI until SBT ranged from 0.04 to 35.2 years, the number of mTBI symptoms in endocrinologist interview ranged from 0 to 6.0, and the number of mTBI symptoms right after concussion from 1.0 to 8.0.

		Total <i>n</i> = 131	No PD <i>n</i> = 115	PD <i>n</i> = 16	Effect size	Р
Sport ^a (%)						
	Soccer	52 (40.3%)	45 (39.8%)	7 (43.8%)	-	-
	Basketball	12 (9.3%)	9 (8.0%)	3 (18.8%)		
	Handball	48 (37.2%)	44 (38.9%)	4 (25.0%)		
	Ice hockey	9 (7.0%)	7 (6.2%)	2 (12.5%)		
	Martial arts	8 (6.2%)	8 (7.1%)	0 (0.0%)		
Still playing (%)						
	Yes	71 (54.2%)	64 (55.7%)	7 (43.8%)	0.08	0.42
	No	60 (45.8%)	51 (44.3%)	9 (56.2%)		
Age (s.d.)		29.3 (7.6)	29.5 (7.7)	27.6 (7.5)	0.25	0.36
BMI (s.d.)		26.3 (4.7)	26.0 (4.5)	28.0 (6.2)	0.42	0.12
SBP (s.d.)		124 (12.1)	124 (12.4)	124 (10.1)	0.00	0.97
DBP (s.d.)		78.4 (9.1)	78.5 (9.1)	77.7 (9.4)	0.09	0.75
Previous medical history (%)						
	Yes	55 (42.0%)	51 (44.3%)	4 (25.0%)	-	-
	No	76 (58.0%)	64 (55.7%)	12 (75.0%)		
Previous hormonal disease (%)						
	Yes	13 (9.9%)	12 (10.4%)	1 (6.3%)	-	-
	No	118 (90.1%)	103 (89.6%)	15 (93.7%)		
HoC (%)						
	No	53 (40.5%)	46 (40.0%)	7 (43.4%)	-	-
	Before mTBI	21 (16.0%)	18 (15.7%)	3 (18.8%)		
	After mTBI	22 (16.8%)	21 (18.3%)	1 (6.3%)		
	Before and after mTBI	35 (26.7%)	30 (26.0%)	5 (3.1%)		
Menstrual changes (%)						
5 ()	Yes, after mTBI	16 (12.2%)	14 (12.2%)	2 (12.5%)	_	-
	No changes	115 (87.8%)	101 (87.8%)	14 (87.5%)		
Years from mTBI (s.p.)	5	5.1 (6.2)	5.2 (6.5)	4.3 (4.2)	0.14	0.58
Number of mTBI (s.p.)		2.2 (0.8)	2.2 (0.7)	2.0 (1.0)	0.23	0.40
Number of mTBI symptoms in endocrinologist interview (s.p.)		2.0 (1.5)	2.1 (1.5)	1.8 (1.4)	0.20	0.52
Number of mTBI symptoms right after concussion (s.d.)		3.4 (1.7)	3.4 (1.6)	3.5 (1.9)	0.06	0.86

^aThe total number of women who answered questions regarding the sport they participated in was *n* = 129, as two women did not report which sport they participated in. Of the 129 women, 113 had PD and 16 did not.

BMI, body mass index; DBP, diastolic blood pressure; HoC, hormonal contraception; mTBI, mild traumatic brain injury; PD, pituitary dysfunction; SBP, systolic blood pressure; s.p., standard deviation.

central hypothyroidism n=4, 3.1%) and 10 (7.5%) had HPRL (prolactinoma n=4, 3.1% and HPRL without prolactinoma n=6, 4.6%). No woman had more than one abnormal hormonal axes. Thus, PD was confirmed in 18.6% of the 86 women referred for further endocrinological evaluation. The mean time from the most recent mTBI until the endocrinological evaluation for the 16 women with PD was 4.3 years (Table 2) (min 2.4 months, max 15.3 years). The six women with HP had a mean time of 4.6 years (min 2.4 months, max 15.3 years) from the most recent mTBI until HP diagnosis was confirmed and the 10 women with HPRL had a mean time of 4.1 years (min 2.4 months, max 11.1 years) until HPRL diagnosis was confirmed. The time from mTBI until the endocrinological work-up for the two women who were diagnosed with GHD was 2 months for one of them and 15 years for the other.

No statistically significant difference was found between the 16 female athletes with PD compared

with those with normal pituitary function concerning prognostic factors (Table 2). The effect size in the current study was largest for age, BMI, the number of mTBI, and the number of mTBI symptoms in the endocrinological interview (0.25, 0.42, 0.23 and 0.20, respectively) (Table 2). The calculated population size needed for 80% power was 261 participants for age, 89 participants for BMI, 296 for the number of mTBI, and 394 or the number of mTBI symptoms in the endocrinological interview. When comparing the number of mTBI symptoms (≤ 3 symptoms or >3symptoms) between the women with HP with those without HP, no statistical significance was found (P=0.19). No statistical significance was found between women with and without HPRL with regards to menstrual disturbances (P = 0.289).

Medical treatment or HST was required for 13 (9.9%, n=131) of the 16 women with PD, thus 81% of women with PD needed medical treatment. All six women diagnosed with HP were started on treatment with HST (levothyroxine for central hypothyroidism n=4,

somatropin for GHD n=2). Seven of the 10 women with HPRL required treatment with a dopamine agonist (cabergoline). Two of these seven women had normal MRI results and five had abnormal MRI results (prolactinoma n=3, no prolactinoma n=2) (Fig 2). Three women with HPRL were asymptomatic and did not need medical treatment (Fig 2). One of them had a pituitary microadenoma and is being followed clinically and the need for treatment reevaluated as necessary.

Discussion

We found that 12.2% (n=131) of female athletes with a history of mTBI had PD (HP 4.6%, HPRL 7.6%). As 50 million people worldwide sustain a TBI annually, with mTBI being the most common (70–90%) (1), this is a very important finding. Moreover, around 1.6 to 3.8 million sport-related mTBI occur annually in the United States (53) and the incidence is likely underestimated (54, 55). This highlights the importance of evaluating pituitary





Figure 2

Results of the lactotroph axis evaluation. Ten women were diagnosed with HPRL, four of them were diagnosed with a prolactinoma. *Both women had elevated S-prolactin levels in SBT and normal MRI results. Both were lost to follow-up, one woman moved abroad and the other has not responded to repeated requests to repeat blood tests for follow-up. **Abnormal MRI without prolactinoma, total count n = 3: symmetrical pituitary enlargement n = 1 without a visible tumor, arachnoid cyst n = 1, symmetrical enlargement of the adenohypophysis without a visible tumor (¥) n = 1. ***Prolactinoma, total count n = 4: microadenoma n = 2, regressing adenoma n = 1, cystic/hemorrhagic adenoma n = 1. ¥ = symmetrical enlargement of the adenohypophysis without a visible tumor. HPRL, hyperprolactinemia.

function following mTBI, especially as symptoms of HP may overlap with mTBI symptoms and HP can be treated. None of the women diagnosed with PD in the current study had ever been evaluated for HP or HPRL despite prominent clinical symptoms following the mTBI which had occurred up to 15.3 years (mean 4.3 years) before the study.

PD was confirmed in 18.6% of the 86 women who had SBT repeatedly O-RV. As 81.4% of the women who were referred for further endocrinological evaluation, did not have PD (45), the question remains which women should be screened and evaluated further for possible PD following mTBI and when should this screening occur? Some studies suggest that increased TBI severity increases the risk of HP (10, 15, 56, 57). However, HP prevalence following mTBI, a more severe brain injury than mTBI, has been reported to be lower than following mTBI (9). Thus, this remains debatable.

We found 4.6%, or six of 131 women, to have HP which is lower than previously reported (13–48%) (8, 9, 10, 11, 12, 13, 14). All six women with HP were treated with HST and experienced symptom relief with treatment during follow-up. This lower prevalence of HP found in our study might be explained by the long interval from the most recent mTBI until the SBT were taken (45). A mean time of 5.1 years passed from the most recent mTBI until the endocrinological evaluation was performed for the entire population (Table 2) and 4.6 years passed from mTBI until the endocrinological evaluation for the six women diagnosed with HP. As it has been suggested that HP may improve with time (33, 58, 59, 60, 61), the extended time interval from mTBI until HP diagnosis may explain the results of our study.

Central hypothyroidism (n=4) was the most common form of HP, followed by GHD (n=2), whereas in previous reports GHD has been most common (9, 10, 12, 13, 14, 15, 33, 36, 62, 63). This is interesting considering the hypothesis that secondary ischemic injury of the pituitary gland may be a possible cause of HP following TBI (16). The TSH- and ACTH-secreting cells reside in the anteromedial portion and the central wedge of the anterior pituitary gland, and the growth hormonesecreting cells reside in the lateral portion of the anterior pituitary gland. As the anteromedial and central wedge of the pituitary gland receives its vasculature from both the long and short hypophyseal portal vessels it should be better guarded from ischemic injury than the lateral portion of the anterior pituitary which receives its vasculature from the hypophyseal portal vessels alone (16, 64, 65, 66).

The prevalence of GHD following mTBI in our study (1.5%, n=131) was lower than the previously reported prevalence of 8–48% (8, 11, 12, 13, 14, 15, 28, 29). Studies have suggested that GHD may improve with time which may explain the lower incidence of GHD in our study. One study found that 53.8% of patients with GHD had recovered after 3 years from TBI, although it is also

discussed that GHD may arise as time passes (59). However, it is interesting that the time from the most recent mTBI until the endocrinological work-up for the two women with GHD was 2 months and 15 years. Thus, one had only recently suffered a mTBI while many years had passed for the other. Another possible explanation for the lower prevalence of GHD is that it may be caused by hypothalamic injury rather than by injury to the pituitary gland itself. As the GHRH–arginine tests was the diagnostic test for GHD rather than the ITT, results may be falsely normal as the GHRH– arginine test does not evaluate possible hypothalamic dysfunction (67, 68, 69).

Ten women (7.6%) had HPRL comparable to previously reported HPRL prevalence of 3.8–16% following mTBI, moTBI and sTBI (15, 31, 32, 33, 34). A mean time of 4.1 years passed from the most recent mTBI until HPRL was diagnosed. Thus, s-prolactin seems to remain elevated for an extended period in some cases following mTBI rather than improving with time as has been suggested for HP (33, 70).

The study did not find hypofunction in the gonadotropic or corticotropic axis (45).

No significant prognostic factors were found (Table 2). Although this is a large study of female athletes after mTBI, our power analysis (Table 2) show that a larger population is needed to identify possible prognostic factors to make PD screening following mTBI more targeted.

Strengths and limitations

To the best of our knowledge, our study is the first to report the prevalence of HP and HPRL following mTBI in an all-female study. This is a strength of the study as female athletes are an underreported population (41). A detailed endocrinological evaluation was conducted by the same endocrinologist and relevant testing were performed when indicated to confirm PD which is also a strength of the study. Furthermore, follow-up with the same endocrinologist was offered to all women who were diagnosed with PD.

It is a limiting factor that the study is retrospective as mTBI tends to be underreported (55, 71, 72, 73). Results from part 1 and part 2 of the study also indicate underreporting as mTBI reporting increased from 40.2% to 64.8% after participants read a definition of mTBI (44). The time from mTBI until the endocrinological evaluation varied between participants which may be a limitation of the study. However, it may also be a strength, as it gives an insight into the long-term prevalence of PD following TBI. Although the study population is larger compared to previous studies on HP following mTBI (8, 9, 10, 11, 12, 13, 14), an even larger study population is needed to identify possible prognostic factors for PD following mTBI. Limitations of the diagnostic methods for possible central cortisol deficiency also need to be considered. Although measurements of s-cortisol and the high-dose (250 μ g) synachten test were performed at 08:00 h, it is possible that variations in individual circadian rhythms could have resulted in false-negative results. Moreover, the use of the synacthen test rather than the gold-standard ITT for the assessment of the hypothalamopituitary–adrenal axis may not be reliable and can lead to false-negative results (74). However, as the ITT requires medical supervision, is physically demanding on patients, and can have contraindications, the synacthen test is often used as an alternative (75).

Conclusions

We conclude that PD is an important diagnosis in postconcussion care as 12.2% of the female athletes (n=16) had PD with 9.9% (n=13) requiring medical treatment. Following mTBI 4.6% of the female athletes had HP which is lower than previously reported and may possibly be explained by recovery over time. HPRL may indicate pituitary or hypothalamic injury after mTBI as six of the 10 women with HPRL were not diagnosed with a prolactinoma.

Declaration of interest

Sigrún Helga Lund is a statistician employed by deCODE genetics Inc./ Amgen Inc.; she is not reimbursed for her work on this study. The authors have no competing interests to declare that are relevant to the content of this article.

Funding

This study was funded by The Icelandic Centre for Research (grant number 207323-052) and Landspitali University Hospital Research Fund.

Compliance with ethical standards

All procedures performed in studies involving human participants were in accordance with the ethical standards of the institutional and/or national research committee and with the 1964 Helsinki Declaration and its later amendments or comparable ethical standards. The study was approved by the National Bioethics Committee (no: VSN-18-091), the Icelandic Data Protection Authority, the Institutional Reasearch Commitee of Landspitali University Hospital of Iceland and Laeknasetrid outpatient clinic. Prior to participation, participants received information regarding the study design and gave their informed consent for participation and publication of the study results.

Data availability

As the participants of this study did not give written consent for their research data to be shared, the data is not available.

Acknowledgements

Guðmundur Sigþórsson, MD, at the Department of Clinical Chemistry at Landspitali the National University Hospital of Iceland, for guidance with the description of the analytical methods utilized for the hormonal measurements. Laeknasetrid outpatient clinic for providing facilities and equipment for the medical interview, physical examination, and ECG.

References

- 1 Maas AIR, Menon DK, Adelson PD, Andelic N, Bell MJ, Belli A, Bragge P, Brazinova A, Büki A, Chesnut RM, *et al.* Traumatic brain injury: integrated approaches to improve prevention, clinical care, and research. *Lancet Neurology* 2017 **16** 987–1048. (https://doi. org/10.1016/S1474-4422(17)30371-X)
- 2 Patricios JS, Schneider KJ, Dvorak J, Ahmed OH, Blauwet C, Cantu RC, Davis GA, Echemendia RJ, Makdissi M, McNamee M, *et al.* Consensus statement on concussion in sport: the 6th International Conference on Concussion in Sport-Amsterdam, October 2022. *British Journal of Sports Medicine* 2023 **57** 695–711. (https://doi.org/10.1136/bjsports-2023-106898)
- 3 Silverberg ND, Iverson GL, Cogan A, Dams-O-Connor K, Delmonico R, Graf MJP, Iaccarino MA, Kajankova M, Kamins J, McCulloch KL, *et al.* The American Congress of Rehabilitation Medicine diagnostic criteria for mild traumatic brain injury. *Archives of Physical Medicine and Rehabilitation* 2023 **104** 1343–1355. (https://doi.org/10.1016/j.apmr.2023.03.036)
- 4 Kazl C & Torres A. Definition, classification, and epidemiology of concussion. *Seminars in Pediatric Neurology* 2019 **30** 9–13. (https:// doi.org/10.1016/j.spen.2019.03.003)
- 5 Lefevre-Dognin C, Cogné M, Perdrieau V, Granger A, Heslot C & Azouvi P. Definition and epidemiology of mild traumatic brain injury. *Neuro-Chirurgie* 2021 67 218–221. (https://doi.org/10.1016/j. neuchi.2020.02.002)
- 6 Voormolen DC, Cnossen MC, Polinder S, von Steinbuechel N, Vos PE & Haagsma JA. Divergent classification methods of postconcussion syndrome after mild traumatic brain injury: prevalence rates, risk factors, and functional outcome. *Journal of Neurotrauma* 2018 **35** 1233–1241. (https://doi.org/10.1089/neu.2017.5257)
- 7 Manley G, Gardner AJ, Schneider KJ, Guskiewicz KM, Bailes J, Cantu RC, Castellani RJ, Turner M, Jordan BD, Randolph C, *et al*. A systematic review of potential long-term effects of sport-related concussion. *British Journal of Sports Medicine* 2017 **51** 969–977. (https://doi.org/10.1136/bjsports-2017-097791)
- 8 Giuliano S, Talarico S, Bruno L, Nicoletti FB, Ceccotti C & Belfiore A. Growth hormone deficiency and hypopituitarism in adults after complicated mild traumatic brain injury. *Endocrine* 2017 **58** 115–123. (https://doi.org/10.1007/s12020-016-1183-3)
- 9 Schneider HJ, Kreitschmann-Andermahr I, Ghigo E, Stalla GK & Agha A. Hypothalamopituitary dysfunction following traumatic brain injury and aneurysmal subarachnoid hemorrhage: a systematic review. JAMA 2007 298 1429–1438. (https://doi. org/10.1001/jama.298.12.1429)
- 10 Klose M, Juul A, Poulsgaard L, Kosteljanetz M, Brennum J & Feldt-Rasmussen U. Prevalence and predictive factors of post-traumatic hypopituitarism. *Clinical Endocrinology* 2007 **67** 193–201. (https:// doi.org/10.1111/j.1365-2265.2007.02860.x)
- 11 Kelestimur F, Tanriverdi F, Atmaca H, Unluhizarci K, Selcuklu A & Casanueva FF. Boxing as a sport activity associated with isolated GH deficiency. *Journal of Endocrinological Investigation* 2004 **27** RC28-RC32. (https://doi.org/10.1007/BF03345299)
- 12 Tanriverdi F, Unluhizarci K, Kocyigit I, Tuna IS, Karaca Z, Durak AC, Selcuklu A, Casanueva FF & Kelestimur F. Brief communication: pituitary volume and function in competing and retired male boxers. *Annals of Internal Medicine* 2008 **148** 827–831. (https://doi. org/10.7326/0003-4819-148-11-200806030-00005)
- 13 Tanriverdi F, Unluhizarci K, Coksevim B, Selcuklu A, Casanueva FF & Kelestimur F. Kickboxing sport as a new cause of traumatic brain injury-mediated hypopituitarism. *Clinical Endocrinology* 2007 66 360–366. (https://doi.org/10.1111/j.1365-2265.2006.02737.x)

- 14 Kelly DF, Chaloner C, Evans D, Mathews A, Cohan P, Wang C, Swerdloff R, Sim MS, Lee J, Wright MJ, *et al.* Prevalence of pituitary hormone dysfunction, metabolic syndrome, and impaired quality of life in retired professional football players: a prospective study. *Journal of Neurotrauma* 2014 **31** 1161–1171. (https://doi. org/10.1089/neu.2013.3212)
- 15 Bondanelli M, De Marinis L, Ambrosio MR, Monesi M, Valle D, Zatelli MC, Fusco A, Bianchi A, Farneti M & degli Uberti EC. Occurrence of pituitary dysfunction following traumatic brain injury. *Journal of Neurotrauma* 2004 **21** 685–696. (https://doi. org/10.1089/0897715041269713)
- 16 Gasco V, Cambria V, Bioletto F, Ghigo E & Grottoli S. Traumatic brain injury as frequent cause of hypopituitarism and growth hormone deficiency: epidemiology, diagnosis, and treatment. *Frontiers in Endocrinology* 2021 **12** 634415. (https://doi.org/10.3389/ fendo.2021.634415)
- 17 Kumar A & Loane DJ. Neuroinflammation after traumatic brain injury: opportunities for therapeutic intervention. *Brain, Behavior,* and Immunity 2012 **26** 1191–1201. (https://doi.org/10.1016/j. bbi.2012.06.008)
- 18 Tanriverdi F, De Bellis A, Battaglia M, Bellastella G, Bizzarro A, Sinisi AA, Bellastella A, Unluhizarci K, Selcuklu A, Casanueva FF, et al. Investigation of antihypothalamus and antipituitary antibodies in amateur boxers: is chronic repetitive head traumainduced pituitary dysfunction associated with autoimmunity? European Journal of Endocrinology 2010 162 861–867. (https://doi. org/10.1530/EJE-09-1024)
- 19 Bullock R, Zauner A, Woodward JJ, Myseros J, Choi SC, Ward JD, Marmarou A & Young HF. Factors affecting excitatory amino acid release following severe human head injury. *Journal of Neurosurgery* 1998 **89** 507–518. (https://doi.org/10.3171/jns.1998.89.4.0507)
- 20 Klose M, Watt T, Brennum J & Feldt-Rasmussen U. Posttraumatic hypopituitarism is associated with an unfavorable body composition and lipid profile, and decreased quality of life 12 months after injury. *Journal of Clinical Endocrinology and Metabolism* 2007 **92** 3861–3868. (https://doi.org/10.1210/jc.2007-0901)
- 21 Prodam F, Gasco V, Caputo M, Zavattaro M, Pagano L, Marzullo P, Belcastro S, Busti A, Perino C, Grottoli S, *et al.* Metabolic alterations in patients who develop traumatic brain injury (TBI)-induced hypopituitarism. *Growth Hormone and IGF Research* 2013 **23** 109–113. (https://doi.org/10.1016/j.ghir.2013.04.001)
- 22 Rosén T, Wirén L, Wilhelmsen L, Wiklund I & Bengtsson BA. Decreased psychological well-being in adult patients with growth hormone deficiency. *Clinical Endocrinology* 1994 **40** 111–116. (https://doi.org/10.1111/j.1365-2265.1994.tb02452.x)
- 23 Reed ML, Merriam GR & Kargi AY. Adult growth hormone deficiency – benefits, side effects, and risks of growth hormone replacement. *Frontiers in Endocrinology* 2013 **4** 64. (https://doi. org/10.3389/fendo.2013.00064)
- 24 Verhelst J, Mattsson AF, Luger A, Thunander M, Góth MI, Koltowska-Häggström M & Abs R. Prevalence and characteristics of the metabolic syndrome in 2479 hypopituitary patients with adultonset GH deficiency before GH replacement: a KIMS analysis. *European Journal of Endocrinology* 2011 **165** 881–889. (https://doi. org/10.1530/EJE-11-0599)
- 25 Rosén T, Johannsson G, Johansson JO & Bengtsson BA. Consequences of growth hormone deficiency in adults and the benefits and risks of recombinant human growth hormone treatment: a review paper. *Hormone Research* 1995 **43** 93–99. (https://doi.org/10.1159/000184245)
- 26 Rosén T & Bengtsson BA. Premature mortality due to cardiovascular disease in hypopituitarism. *Lancet* 1990 **336** 285–288. (https://doi.org/10.1016/0140-6736(90)91812-0)

- 27 Biondi B, Cappola AR & Cooper DS. Subclinical hypothyroidism: a review. JAMA 2019 322 153–160. (https://doi.org/10.1001/ jama.2019.9052)
- 28 Lithgow K, Chin A, Debert CT & Kline GA. Utility of serum IGF-1 for diagnosis of growth hormone deficiency following traumatic brain injury and sport-related concussion. *BMC Endocrine Disorders* 2018 **18** 20. (https://doi.org/10.1186/s12902-018-0247-1)
- 29 Ioachimescu AG, Hampstead BM, Moore A, Burgess E & Phillips LS. Growth hormone deficiency after mild combat-related traumatic brain injury. *Pituitary* 2015 **18** 535–541. (https://doi.org/10.1007/ s11102-014-0606-5)
- 30 Sav A, Rotondo F, Syro LV, Serna CA & Kovacs K. Pituitary pathology in traumatic brain injury: a review. *Pituitary* 2019 **22** 201–211. (https://doi.org/10.1007/s11102-019-00958-8)
- 31 Agha A, Rogers B, Sherlock M, O'Kelly P, Tormey W, Phillips J & Thompson CJ. Anterior pituitary dysfunction in survivors of traumatic brain injury. *Journal of Clinical Endocrinology and Metabolism* 2004 **89** 4929–4936. (https://doi.org/10.1210/jc.2004-0511)
- 32 Bensalah M, Donaldson M, Labassen M, Cherfi L, Nebbal M, Haffaf EM, Abdennebi B, Guenane K, Kemali Z & Ould Kablia S. Prevalence of hypopituitarism and quality of life in survivors of post-traumatic brain injury. *Endocrinology, Diabetes and Metabolism* 2020 **3** e00146. (https://doi.org/10.1002/edm2.146)
- 33 Aimaretti G, Ambrosio MR, Di Somma C, Gasperi M, Cannavo S, Scaroni C, Fusco A, Del Monte P, De Menis E, Faustini-Fustini M, et al. Residual pituitary function after brain injury-induced hypopituitarism: a prospective 12-month study. Journal of Clinical Endocrinology and Metabolism 2005 90 6085–6092. (https://doi. org/10.1210/jc.2005-0504)
- 34 Kokshoorn NE, Wassenaar MJ, Biermasz NR, Roelfsema F, Smit JW, Romijn JA & Pereira AM. Hypopituitarism following traumatic brain injury: prevalence is affected by the use of different dynamic tests and different normal values. *European Journal of Endocrinology* 2010 **162** 11–18. (https://doi.org/10.1530/EJE-09-0601)
- 35 Olivecrona Z, Dahlqvist P & Koskinen LO. Acute neuro-endocrine profile and prediction of outcome after severe brain injury. *Scandinavian Journal of Trauma, Resuscitation and Emergency Medicine* 2013 **21** 33. (https://doi.org/10.1186/1757-7241-21-33)
- 36 Aimaretti G, Ambrosio MR, Di Somma C, Fusco A, Cannavò S, Gasperi M, Scaroni C, De Marinis L, Benvenga S, degli Uberti EC, *et al.* Traumatic brain injury and subarachnoid haemorrhage are conditions at high risk for hypopituitarism: screening study at 3 months after the brain injury. *Clinical Endocrinology* 2004 **61** 320–326. (https://doi.org/10.1111/j.1365-2265.2004.02094.x)
- 37 Agha A, Rogers B, Mylotte D, Taleb F, Tormey W, Phillips J & Thompson CJ. Neuroendocrine dysfunction in the acute phase of traumatic brain injury. *Clinical Endocrinology* 2004 **60** 584–591. (https://doi.org/10.1111/j.1365-2265.2004.02023.x)
- 38 Bornstein SR, Allolio B, Arlt W, Barthel A, Don-Wauchope A, Hammer GD, Husebye ES, Merke DP, Murad MH, Stratakis CA, et al. Diagnosis and treatment of primary adrenal insufficiency: an Endocrine Society clinical practice guideline. Journal of Clinical Endocrinology and Metabolism 2016 101 364–389. (https://doi. org/10.1210/jc.2015-1710)
- 39 Tritos NA & Biller BMK. Current concepts of the diagnosis of adult growth hormone deficiency. *Reviews in Endocrine and Metabolic Disorders* 2021 **22** 109–116. (https://doi.org/10.1007/s11154-020-09594-1)
- 40 Ho KKY. Consensus guidelines for the diagnosis and treatment of adults with GH deficiency II: a statement of the GH Research Society in association with the European Society for Pediatric Endocrinology, Lawson Wilkins Society, European Society of

Endocrinology, Japan Endocrine Society, and Endocrine Society of Australia. *European Journal of Endocrinology* 2007 **157** 695–700. (https://doi.org/10.1530/EJE-07-0631)

- 41 D'Lauro C, Jones ER, Swope LM, Anderson MN, Broglio S & Schmidt JD. Under-representation of female athletes in research informing influential concussion consensus and position statements: an evidence review and synthesis. *British Journal of Sports Medicine* 2022 **56** 981–987. (https://doi.org/10.1136/ bisports-2021-105045)
- 42 Covassin T, Swanik CB & Sachs ML. Sex differences and the incidence of concussions among collegiate athletes. *Journal of Athletic Training* 2003 **38** 238–244.
- 43 McGroarty NK, Brown SM & Mulcahey MK. Sport-related concussion in female athletes: a systematic review. Orthopaedic Journal of Sports Medicine 2020 8 2325967120932306. (https://doi. org/10.1177/2325967120932306)
- 44 Kristjánsdóttir H, Brynjarsdóttir RM, Kristensen ISU, Sigurjónsdóttir HÁ, Claessen LÓE & Jónsdóttir MK. Self-reported concussion history among Icelandic female athletes with and without a definition of concussion. *Clinical Neuropsychologist* 2020 34(sup1) 70–82. (https://doi.org/10.1080/13854046.2020.1814873)
- 45 Claessen LÓE, Kristjánsdóttir H, Jónsdóttir MK, Lund SH, Kristensen ISU & Sigurjónsdóttir HÁ. Screening for possible hypopituitarism following mild traumatic brain injury: the first allfemale study. Who do we need to evaluate further? *NeuroRehabilitation* 2023 **52** 259–271. (https://doi.org/10.3233/NRE-220194)
- 46 Jónsdóttir MK, Kristófersdóttir KH, Runólfsdóttir S, Kristensen ISU, Sigurjónsdóttir HÁ, Eggertsdóttir Claessen LÓ & Kristjánsdóttir H. Concussion among female athletes in Iceland: stress, depression, anxiety, and quality of life. *Nordic Psychology* 2022 **74** 262–278. (https://doi.org/10.1080/19012276.2021.2004916)
- 47 Oertel H, Schneider HJ, Stalla GK, Holsboer F & Zihl J. The effect of growth hormone substitution on cognitive performance in adult patients with hypopituitarism. *Psychoneuroendocrinology* 2004 **29** 839–850. (https://doi.org/10.1016/S0306-4530(03)00151-3)
- 48 Deijen JB, de Boer H, Blok GJ & van der Veen EA. Cognitive impairments and mood disturbances in growth hormone deficient men. *Psychoneuroendocrinology* 1996 **21** 313–322. (https://doi. org/10.1016/0306-4530(95)00050-x)
- 49 Manosroi W, Phimphilai M, Khorana J & Atthakomol P. Diagnostic performance of basal cortisol level at 0900–1300h in adrenal insufficiency. *PLoS One* 2019 **14** e0225255. (https://doi.org/10.1371/ journal.pone.0225255)
- 50 Dorin RI, Qualls CR & Crapo LM. Diagnosis of adrenal insufficiency. Annals of Internal Medicine 2003 139 194–204. (https://doi. org/10.7326/0003-4819-139-3-200308050-00009)
- 51 Wallace I, Cunningham S & Lindsay J. The diagnosis and investigation of adrenal insufficiency in adults. *Annals of Clinical Biochemistry* 2009 **46** 351–367. (https://doi.org/10.1258/ acb.2009.009101)
- 52 Torre DL & Falorni A. Pharmacological causes of hyperprolactinemia. *Therapeutics and Clinical Risk Management* 2007 **3** 929–951.
- 53 Langlois JA, Rutland-Brown W & Wald MM. The epidemiology and impact of traumatic brain injury: a brief overview. *Journal of Head Trauma Rehabilitation* 2006 **21** 375–378. (https://doi. org/10.1097/00001199-200609000-00001)
- 54 McCrea M, Hammeke T, Olsen G, Leo P & Guskiewicz K. Unreported concussion in high school football players: implications

for prevention. *Clinical Journal of Sport Medicine* 2004 **14** 13–17. (https://doi.org/10.1097/00042752-200401000-00003)

- 55 McDonald T, Burghart MA & Nazir N. Underreporting of concussions and concussion-like symptoms in female high school athletes. *Journal of Trauma Nursing* 2016 23 241–246. (https://doi. org/10.1097/JTN.0000000000227)
- 56 Glynn N & Agha A. Which patient requires neuroendocrine assessment following traumatic brain injury, when and how? *Clinical Endocrinology* 2013 **78** 17–20. (https://doi.org/10.1111/ cen.12010)
- 57 Kgosidialwa O & Agha A. Hypopituitarism post traumatic brain injury (TBI): review. *Irish Journal of Medical Science* 2019 **188** 1201–1206. (https://doi.org/10.1007/s11845-019-02007-6)
- 58 Schneider HJ, Schneider M, Saller B, Petersenn S, Uhr M, Husemann B, von Rosen F & Stalla GK. Prevalence of anterior pituitary insufficiency 3 and 12 months after traumatic brain injury. *European Journal of Endocrinology* 2006 **154** 259–265. (https://doi. org/10.1530/eje.1.02071)
- 59 Tanriverdi F, Ulutabanca H, Unluhizarci K, Selcuklu A, Casanueva FF & Kelestimur F. Three years prospective investigation of anterior pituitary function after traumatic brain injury: a pilot study. *Clinical Endocrinology* 2008 68 573–579. (https://doi. org/10.1111/j.1365-2265.2007.03070.x)
- 60 Klose M, Juul A, Struck J, Morgenthaler NG, Kosteljanetz M & Feldt-Rasmussen U. Acute and long-term pituitary insufficiency in traumatic brain injury: a prospective single-centre study. *Clinical Endocrinology* 2007 67 598–606. (https://doi. org/10.1111/j.1365-2265.2007.02931.x)
- 61 Krahulik D, Zapletalova J, Frysak Z & Vaverka M. Dysfunction of hypothalamic-hypophysial axis after traumatic brain injury in adults. *Journal of Neurosurgery* 2010 **113** 581–584. (https://doi. org/10.3171/2009.10.JNS09930)
- 62 Yaseen NT, Al-Khaqani FA & Mansour AA. Patterns of pituitary dysfunction three months or more after traumatic brain injury. *Avicenna Journal of Medicine* 2018 **8** 125–132. (https://doi. org/10.4103/ajm.AJM_2_18)
- 63 Tanriverdi F, Senyurek H, Unluhizarci K, Selcuklu A, Casanueva FF & Kelestimur F. High risk of hypopituitarism after traumatic brain injury: a prospective investigation of anterior pituitary function in the acute phase and 12 months after trauma. *Journal of Clinical Endocrinology and Metabolism* 2006 **91** 2105–2111. (https://doi.org/10.1210/jc.2005-2476)
- 64 Xuereb GP, Prichard MM & Daniel PM. The arterial supply and venous drainage of the human hypophysis cerebri. *Quarterly Journal of Experimental Physiology and Cognate Medical Sciences* 1954 **39** 199–217. (https://doi.org/10.1113/expphysiol.1954. sp001072)
- 65 Xuereb GP, Prichard ML & Daniel PM. The hypophysial portal system of vessels in man. *Quarterly Journal of Experimental Physiology and Cognate Medical Sciences* 1954 **39** 219–230. (https:// doi.org/10.1113/expphysiol.1954.sp001073)
- 66 Gorczyca W & Hardy J. Arterial supply of the human anterior pituitary gland. *Neurosurgery* 1987 **20** 369–378. (https://doi. org/10.1227/00006123-198703000-00003)
- 67 Darzy KH, Aimaretti G, Wieringa G, Gattamaneni HR, Ghigo E & Shalet SM. The usefulness of the combined growth hormone (GH)releasing hormone and arginine stimulation test in the diagnosis of radiation-induced GH deficiency is dependent on the postirradiation time interval. *Journal of Clinical Endocrinology and Metabolism* 2003 **88** 95–102. (https://doi.org/10.1210/jc.2002-021094)

- 68 Glynn N & Agha A. Diagnosing growth hormone deficiency in adults. *International Journal of Endocrinology* 2012 2012 972617. (https://doi.org/10.1155/2012/972617)
- 69 Schneider HJ, Herrmann BL, Schneider M, Sievers C, Schaaf L & Stalla GK. Discrepant results in the diagnosis of GH deficiency with the insulin-tolerance test and the GHRH plus arginine test in patients with traumatic brain injury. *European Journal of Endocrinology* 2006 **155** 553–557. (https://doi.org/10.1530/ eje.1.02258)
- 70 Giordano G, Aimaretti G & Ghigo E. Variations of pituitary function over time after brain injuries: the lesson from a prospective study. *Pituitary* 2005 8 227–231. (https://doi.org/10.1007/s11102-006-6045-1)
- 71 Kerr ZY, Register-Mihalik JK, Kroshus E, Baugh CM & Marshall SW. Motivations associated with nondisclosure of self-reported concussions in former collegiate athletes. *American Journal of Sports Medicine* 2016 **44** 220–225. (https://doi. org/10.1177/0363546515612082)

- 72 Register-Mihalik JK, Guskiewicz KM, McLeod TCV, Linnan LA, Mueller FO & Marshall SW. Knowledge, attitude, and concussionreporting behaviors among high school athletes: a preliminary study. *Journal of Athletic Training* 2013 48 645–653. (https://doi. org/10.4085/1062-6050-48.3.20)
- 73 Anderson BL, Gittelman MA, Mann JK, Cyriac RL & Pomerantz WJ. High school football players' knowledge and attitudes about concussions. *Clinical Journal of Sport Medicine* 2016 26 206–209. (https://doi.org/10.1097/JSM.0000000000214)
- 74 Ammari F, Issa BG, Millward E & Scanion MF. A comparison between short ACTH and insulin stress tests for assessing hypothalamo-pituitary-adrenal function. *Clinical Endocrinology* 1996 44 473–476. (https://doi.org/10.1046/j.1365-2265.1996.712533.x)
- 75 Cho HY, Kim JH, Kim SW, Shin CS, Park KS, Kim SW, Jang HC & Kim SY. Different cut-off values of the insulin tolerance test, the high-dose short Synacthen test (250 μg) and the low-dose short Synacthen test (1 μg) in assessing central adrenal insufficiency. *Clinical Endocrinology* 2014 **81** 77–84. (https://doi.org/10.1111/cen.12397)